Author's response to reviews

Title: Accidental exposure to gas emissions from transit goods treated for pest control.

Authors:

Stefan Kloth (stefan.kloth@googlemail.com)
Xaver Baur (xaver.baur@charite.de)
Thomas Göen (thomas.goeven@ipasum.med.uni-erlangen.de)
Lygia T Budnik (L.budnik@uke.de)

Version: 2
Date: 1 August 2014

Author's response to reviews:

As suggested we have included the missing part: Author contribution (see revised Version of the manuscript)